### Accepted Manuscript

Empirical investigation of key business factors for digital game performance

Saiqa Aleem, Luiz Fernando Capretz, Faheem Ahmed

| PII:           | S1875-9521(15)00013-0                          |
|----------------|--|
| DOI:           | http://dx.doi.org/10.1016/j.entcom.2015.09.001 |
| Reference:     | ENTCOM 157                                     |
| To appear in:  | Entertainment Computing                        |
| Received Date: | 21 January 2015                                |
| Revised Date:  | 10 June 2015                                   |
| Accepted Date: | 16 September 2015                              |



Please cite this article as: S. Aleem, L.F. Capretz, F. Ahmed, Empirical investigation of key business factors for digital game performance, *Entertainment Computing* (2015), doi: http://dx.doi.org/10.1016/j.entcom.2015.09.001

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

# Empirical investigation of key business factors for digital game performance

#### Saiqa Aleem<sup>a,\*</sup>, Luiz Fernando Capretz<sup>a</sup>, Faheem Ahmed<sup>b</sup>

<sup>a</sup> Department of Electrical & Computer Engineering, University of Western Ontario, London, Ontario, Canada N6A 5B9 <sup>b</sup> Department of Computing Science, Thompson Rivers University, Kamloops, British Columbia, Canada, V2C 0C8

#### ARTICLE INFO

Article history: Received Received in revised form Accepted

Keywords: Software game Game business performance Video game Online game Empirical Investigation Games business factors

\* Corresponding author.

#### E-mail addresses: saleem4@uwo.ca

Phone number: +1-226-977-0417; Fax number: +1-519-850-2436.

#### ABSTRACT

Game development is an interdisciplinary concept that embraces software engineering, business, management, and artistic disciplines. This research facilitates a better understanding of the business dimension of digital games. The main objective of this research is to investigate empirically the effect of business factors on the performance of digital games in the market and to answer the research questions asked in this study. Game development organizations are facing high pressure and competition in the digital game industry. Business has become a crucial dimension, especially for game development organizations. The main contribution of this paper is to investigate empirically the influence of key business factors on the business performance of games. This is the first study in the domain of game development that demonstrates the interrelationship between key business factors and game performance in the market. The results of the study provide evidence that game development organizations must deal with multiple business key factors to remain competitive and handle the high pressure in the digital game industry. Furthermore, the results of the study support the theoretical assertion that key business factors play an important role in game business performance.

#### 1. Introduction

Over the past three decades, digital games have emerged as an important part of media and global entertainment. The digital game sector is creative, dynamic, pervasive, and exciting. The social media revolution and ever-increasing Internet expansion are driving phenomenal growth for the digital game segment in particular and are creating a huge multimedia business worth billions of dollars. The digital game sector, especially video games, is expected to grow by up to USD 112 billion in sales by 2015 as reported by Gartner Research [1], and overall growth of the digital game sector is expected to reach USD 82.4 billion by 2015 [2]. Digital game development organizations are looking at new ways to improve existing user experiences, to engage a broad range of consumers, to update their business models, and to include emerging technologies in their development processes. The digital game sector has been identified as a significant contributor to economic growth by many countries around the world, and these countries have embraced aggressive action plans for its growth [3, 4]. In the digital game industry, Kerr [5] identified four distinct segments: standard PC, console, casual, and massively multiplayer online games. Organizations in each segment have a different culture for production and entry to market and are structured differently. Game development organizations are directly or indirectly involved in various activities from a game's inception to its consumption. These main activities in general terms, regardless of game genre or particular segment, can be grouped together as: a) development or production; b) commercialization or publishing; c) distribution; and d) customer engagement. Production or development is a major multidisciplinary activity that involves merging of technical and creative disciplines. The development process involves planning, design, development, and test phases and is a kind of iterative process. The publishing activity involves either developing games in-house and outsourcing publishing of developed games or else purchasing of games from independent studios by publisher companies. Outsourced publishing activities can include data hosting, billing, marketing and advertising, intellectual property management, and analytics. Distribution activity is different for online and packaged games. For online games, intermediaries like virtual windows can be used for their distribution. For packaged games, distribution includes handling and packaging of games as well as marketing and logistics. Distribution activities can also be outsourced by game companies. Customer engagement includes customer support activity; some companies also outsource customer support to achieve cost savings.

However, the main development activities in the digital game sector especially from business perspective include elements of the game development value chain, such as technical and creative development, manufacturing of hardware/ console platforms, and game publishing. Distribution can be carried out in a number of ways, including mobile, traditional retail, online, cloud, and download, and after distribution, it also involves customer engagement and community management activities. Game development organizations have also outsourced some of their support services activities such as data hosting, information security, marketing and advertising, billing services, and piracy protection. The number of dimensions involved, such as types of end-user devices or platforms, game genres, channels for publication, and emerging revenue models in the digital game sector, make this sector highly fragmented. It is important for any type of business domain to identify its key important factors that help them to excel in that particular field. The key business factors vary from domain to domain depending upon their business operation. Digital game is kind of software product and it is intangible in nature. According to Levitt [6], intangible products are highly people-intensive in their delivery methods and production and business management become more critical for them as compared to tangible products. Moreover, digital game industry growth is tremendous and it became crucial to identify key important business factors that help organizations in digital game industry to reach their maximum potential. Game development organizations must target all these dimensions to retain and maximize their consumers. The digital game industry has shown economic potential in both the entertainment and software industries [2].

#### 1.1 Research Motivation

Organizations involved in the game development business are facing stiff competition and high consumer expectations because more and more development companies are entering the digital game industry day by day. The main research motivation behind this study is the rapid and continual changes in technology and the severity of competition in game

development organization and it not only affect the business, but also have a great impact on development activities. Actually, the game industry has high economic potential and generates million-dollar projects, it sets high limits and standards for game performance as well as putting high pressure on organizations. To deal with this severe competition and high pressure, game development organizations must make important decisions quickly regarding different business activities because this has become important for financial growth and business performance. Organizations in the digital game industry must respond quickly to changes in the business and technology environment, and if they fail to respond appropriately, then they will not survive long. There are many examples of commercial failure in digital game industry and the most popular one is known as video game crash of 1983 [7]. According to Burnham [7], an expensive low quality games with poor business strategies were flooded in North America. They resulted in complete destruction of US digital game market. Also, Sellers [8] stated that the extra-terrestrial (E.T.) video game and Pac Man for Atrai 2600 were two examples that contributed to the failure. Most of the failures in the digital game market such as Commodore 64 Games System, Nitendo 64 DD, Philips Cdi, Shenmue, Sonic Boom: Rise of lyric etc. [9], were due to poor business strategies including market orientation, consumer satisfaction, monetization strategy, time to market etc.

Especially in game development organizations, business becomes the most important factor due to severe competition, the fragmented nature of the business, and the poor software engineering practices used by most companies [5]. Identification of key factors to handle high pressure and achieve targeted business and game performance has become highly important. However, no studies that address the important factors in digital game business performance have been published in the literature. The main contribution of this empirical study is to investigate comprehensively the interrelationship among key business factors and game performance in the market. This study also provides an understanding of the influence of the key factors identified by showing empirically how they impact the business organization and digital game performance.

The rest of the paper is organized as follows: Section 2 provides the research methodology, and Section 3 describes the results and analysis. Section 4 presents a discussion, and Section 5 concludes the study.

#### 2. Literature Review

Key business performance factors for digital game organizations are the least addressed area in game development research. The business model for each segment of the game industry is different, and each segment has a different percentage of the revenue share [10]. From a review of the literature, various factors have been identified that contribute to game business performance. The identified factors and the related literature review are described in the following subsections.

#### 2.1 Customer Satisfaction/Loyalty

The digital game industry (DGI) is facing dramatic changes because it views customer satisfaction as winning over players for their games. The classical definition of customer satisfaction given by Oliver [11] is "pleasurable fulfillment response toward a good, service, benefit, or reward". Customer satisfaction must be an integral part of the organization and is a financial metric that can be used to measure business performance. However, the relationship between business performance and customer satisfaction is not always clear. Zeithaml [12] highlighted three problems in measuring this relationship: a) the time lag between measuring improvement in profit and customer satisfaction; b) other variables that influence an organization's profits, such as marketing, price, and competition; and c) other variables such as organizational behavioral issues that should be included when measuring the relationship. A positive relationship between customer satisfaction in any organization is strongly correlated with higher market growth, proving the strong relationship between customer loyalty and customer retention. The DGI has given a lower priority to customer service for its product (the game) and tends to treat it as a commodity. Often, when players do not obtain a response to their problems, they become disappointed. Johnson [21] explored the aspect of customer service in the DGI. He used the critical incidents technique to examine customer services incidents in the game industry and identified negative and positive customer

service experiences. The results of this study provided directions for management that helped them with resource allocation, especially in those areas that provided maximum customer satisfaction and dissatisfaction. Based on this analysis, management could take proper measures to ensure maximum customer satisfaction. In commercial games, the concept of customer satisfaction has a very important place. Lu and Wang [22] explored the factors of online game addiction and the role of addiction in online gamer loyalty and customer satisfaction. The results indicated that addiction plays an important role in customer loyalty and satisfaction. Wiele *et al.* [23] investigated the relationship of customer satisfaction and business performance data within an organization. The results showed empirical evidence that there is a positive relationship between customer satisfaction and business performance. In the literature, only a few researchers have explored the customer satisfaction aspect of the DGI.

#### 2.2 Market Orientation

Market orientation plays a significantly important role in extensively market-driven DGI. Market orientation involves the study of customers and competitors in the market and deals with the interpretation, acquisition, or use of information about them. The concept of market orientation is based on marketing theory. Zeithaml and Zeithaml's [24] marketing theory also applies here because it provides continuous guidance for game development organizations on how they should react to opportunities and how, by taking appropriate market actions, the organization can create opportunities by changing the environment. Hunt [25] describes marketing as a management responsibility that helps in sensing the market and articulating new and valuable propositions. Berry [26] also highlighted the use of customer relationship management (CRM) to develop an appropriate marketing strategy to retain, attract, and enhance customer relationships. Gronroos [27] defined marketing in the context of CRM, and Fornell and Wernerfelt [28] described a marketing strategy aimed at attracting new and retaining existing customers and resulting in increased revenue and profitability. Owomoyela *et al.* [29] described how organizations can develop their marketing strategies in a way that enables them to build, maintain, and defend their competitive advantage. Managerial judgment will be helpful in identifying strategic marketing uncertainties and environmental ambiguities.

In the literature, very few studies have described market orientation for the DGI. Lee *et al.* [30] suggested that game developers must develop market reports in the requirements engineering phase and during game distribution. The marketing team plays an important role in this. The main activities performed by marketing teams along with the CRM team are packaging, advertising, management of marketing agents, and production of a complete marketing plan. Kasaliaki and Mustafee [31] explored sustainable development strategies for a serious game audience. Analyzing the characteristics and requirements of the target audience helps developers generate a sustainable game development process. Xin [32] highlighted the barriers in serious mobile game markets and the current market segmentation for serious games. Before developing serious games, developers must analyze the market segment and their own competitive advantage. This study highlights the issue of market analysis before starting a game project to determine what types of games are in demand.

#### 2.3 Innovation

Especially in the DGI, innovation has a special place as a key driver of economic growth and competitiveness. Innovation has many forms and has become known as a critical dimension of achieving better economic performance, especially in knowledge-driven economies. Innovation can be defined as the successful exploitation of new social or commercial ideas and the ability, once new ideas have been brought to market, to reduce cost, improve services, and improve existing arrangements by offering new and effective alternatives. Afuah [33] defined strategic innovation for organizations as follows: "a strategic innovation is a game-changing innovation in products/services, business models, business processes, and/or positioning of competitors to improve performance". Johannessen [34] described a systematic perspective on innovation theory. He considered 14 propositions from the literature and investigated the connection between economic crises and innovation. He categorized innovation into two major categories: institutional and economic innovation. Furthermore, institutional innovations were categorized into political, cultural, and social innovations. The economic innovations in organizations are associated with managing an organization in new ways as well as with new business models. A business model innovation framework has been proposed by Comviva Technologies [35] that contains an industry model (adoption of new industries by redefining existing ones), a revenue model (reconfiguration of offerings and a pricing model), and an

expertise model (value-chain role playing). Lindgardth *et al.* [36] also proposed an innovative business model including two elements: a value proposition and an operating model. The value proposition is about who the target audience is, what kind of product/service the organization will offer, and what the organizational revenue model will be. The operating model addresses the issue of service/product delivery that generates profitability and includes three critical areas: the value chain, a cost model to generate revenue, and an organization that develops and deploys assets to enhance and sustain competitive advantage.

#### 2.4 Relationship Management

Effective CRM is a highly critical element in the success of any business. Wilson [37] observed that relationship management basically involves developing and maintaining long-term, close, satisfying and mutually beneficial relationships between customers and organizations based on collaboration and trust. In relationship management, customer profiling, promotional strategies, customer service and support, customer information, organizational behavior, and channel management are all contributing factors. Recently, organizations have been integrating their customers into the design, production, or delivery of goods and services. These organizations are mainly targeting revenue increases or cost reductions by relying on their customers as co-producers of goods or services that they offer to the market. This trend towards integrating users or customers shows that new organizational choices are being made by companies to generate high margins. This is a fundamental change in business strategy that pushes organizations to think about new ways to mobilize their users to increase revenue. Plé et al. [38] explained the role of customers in this business model. They proposed a theoretical framework called the Customer-Integrated Business Model (CIBM) by combining customer participation with the business model literature. The framework based on ROCA (Resource Oriented Client Architecture) and proposed by Lecocq et al. [39] considered the customer as a resource; the model was illustrated by two case studies. They concluded that more field research is required to explore the relationship between the customer-as-a-resource approach and business profits. Most studies in the literature consider customer participation in service marketing and management, whereas only a few consider customer integration as a resource. The digital game literature also lacks the dimension of customer integration for business performance. Stanely et al. [40] looked at user integration from a different perspective. They described the cumulative context of a digital game and accumulated all contextual information on a player's activity using mobile sensors to change the game state. Experimental results indicated that the player found the game engaging and fun. Ermi and Mayra [41] pointed out that user involvement is a multi-dimensional and complex phenomenon which is not totally dependent on the nature of the specific genre or game, but also upon each player's choices or preferences.

#### 2.5 Time to Market

The time-to-market phenomenon has long been recognized as a crucial enabler for business success. From this perspective, organizations can be categorized into pioneers, early followers, and late movers [42, 43]. The pioneers emerge as solution providers in the market and gain a sustainable competitive advantage over followers. This enables them to amass a major part of the market, making it more difficult for successors to gain market share. Hence, the timing of entry into the market becomes more crucial for organizations to gain profit and competitive advantage. Products that enter the market at the right time or have short time to market have high potential of success. A digital game organization's ability to reach the market before their competitors and gain adoption is an important factor in the long-term success of games. The time-to-market process in the DGI can be defined as integration of new technology into digital game production. Today, digital game organizations can gain competitive advantage by introducing the next generation of technologies into the game market through new game development strategies that enable them to be first in the game industry market. Very few studies in the literature have highlighted the importance of the time-to market factor specifically in the digital game industry [44,30], and none of them has discussed it from a business performance perspective.

#### 2.6 Monetization Strategy

The DGI sector is learning the game of monetization. Around the world, millions of consumers play games on either online media portals or social networking sites every month. Monetization strategy is very important because it is a risky business. It provides an insight into the organization of a business that either is worthwhile or is not. Monetization strategy in games is similar to the setting of financial objectives for any organization. Financial objectives are defined as organizations set their financial targets over a certain period of time. Financial objectives are different from other types of organizational

objectives such as business or customer retention objectives because they cannot be easily measured monetarily if achieved. Game development involves high costs, and only the top 5% of games in the market are profitable. A game that fails in the market can lead to severe losses or even bankruptcy in the case of small developers. The organization needs therefore to have proper financial management and appropriate financial planning to ensure that enough funding is available when needed. Second, financial controls determine whether the organization is meeting its financial objectives. Finally, financial decision-making is itself very important [45].

In social games, players are able to create their own virtual characters and communities and interact with their friends. Companies involved in the game business have developed business models for paid content such as subscription, advertising, and micro-transactions for virtual goods. In general, users are not interested in paying for virtual goods, but the few who pay for them make this business model work. Eventually, micro-transactions, especially in the social game lifecycle, have become a driver for incremental revenue. In the massive multiplayer online game sector (MMO), the bulk of game revenue is still generated by subscriptions, but use of micro-transactions is growing for virtual goods. The importance of a monetization strategy for the DGI has been explored by only two studies, but not in detail, and neither of them discussed its impact on business performance [46].

#### 2.7 Brand Name Strategy

A brand name is regarded as a crucial enabler for business success in any organization. The brand is considered as both a point of comparison with other products and a promise of quality to the customer. Bennett [47] described a brand as a term, name, symbol, sign, design, or combination of any of these concepts that helps to identify the products or services of a particular seller. Generally, the brand name has high impact on the organization's business. Between the organization and its customers, branded products serve as an interface, and brand loyalty enables marketing by word of mouth. The organization's brand name strategy has a strong impact on the customer decision-making process. Bergstrom [48] perceived that in the case of products and competitors that are easily replicable or duplicated, brands help customers in the decision process of buying a particular product.

Hence, the DGI has successfully adopted a brand name strategy in the game development process. In games, there are many successful platform brands, including Nintendo, Sony, and Microsoft for consoles, Apple (IOS), Samsung (Android), and others for mobile platforms, and Windows, Apple, and others for PCs. However, no study has described the brand name strategy in the game development process and its impact on business performance.

#### 3. Research Model and Hypotheses

The main objective of the proposed research model is to analyze the interrelationship between key factors and game business performance and also to understand the influence of these factors on a game development organization's business performance in the DGI market. Davenport [49] and Aguilar-Sav'en [50] described the combination of structured business process activities in an organization to achieve specific goals. The model's theoretical foundation is based on a combination of existing concepts found in the game development literature and business models for the game industry. It is worth noting that most studies in the literature discuss one or two of the factors mentioned above in the context of game development organizations and their impact on game performance. To the best of the authors' knowledge, this is the first study in the game development literature that highlights key factors for game performance in game development organizations and game business performance. The Fig. 1 presented the theoretical research model of this study to be empirically investigated. The theoretical model evaluates the relationships of different independent variables emerging from organizational concepts such as organizational management, theory, and behavior in the context of game development organizations on the dependent variable of game business performance within the organization. This study mainly investigates and addresses the following research question:

Research Question: What is the impact of key business factors on overall game business performance in the DGI?

6



Fig 1. Research model

The research model includes seven independent variables: customer satisfaction, market orientation, innovation, relationship management, time to market, monetization strategy, and brand name strategy, and one dependent variable: the business performance of the digital game. The multiple linear regression equation of the model is given as Equation 1: Business performance of game =  $\beta_0 + \beta_1 f_1 + \beta_2 f_2 + \beta_3 f_3 + \beta_4 f_4 + \beta_5 f_5 + \beta_6 f_6 + \beta_7 f_7$ , (1)

where  $\beta_0$ ,  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$ ,  $\beta_4$ ,  $\beta_5$ ,  $\beta_6$ ,  $\beta_7$  are coefficients and  $f_1$ - $f_7$  are the seven independent variables. For purposes of empirical investigation, the following hypotheses are stated:

H1: Customer satisfaction has a positive impact on the performance of a digital game.

H2: Market orientation has a positive impact on the performance of a digital game.

H3: Innovation has a positive impact on the performance of a digital game.

H4: Relationship management has a positive impact on the performance of a digital game.

**H5:** Time to market has a positive impact on the performance of a digital game.

H6: Monetization strategy has a positive impact on the performance of a digital game.

H7: Brand name strategy has a positive impact on the performance of a digital game

#### 4. Research Methodology

Digital game development organizations are involved in various business activities such as game development, publishing, distribution, and finally customer engagement. The targeted respondents of this study were employees of game development organizations or independent studios. Some organizations handled all these activities by themselves, whereas some of them outsourced publishing or distribution activities. Initially, the authors joined various game development community forums and started blogs about a data collection request for an empirical study. A survey questionnaire was also created using the Survey Monkey Web site, and personalized emails were sent to various organizations. The respondents

were from multinational companies in Asia, Europe, and North America, and statistics about them are illustrated in Fig. 2. Participant organizations agreed to take part in the study based on mutual agreement that their identities would be kept confidential. The size of the participating organizations varied from micro to large scale. Micro size organizations consisted of 3–5, small ones of 5–99, medium ones of 100–499, and large ones of 500+ team members belonging to various departments within the organization. Figure 3 show the number of respondents by organization size.



Fig. 3 Number of respondents by organization size

The participating organizations mainly developed games for different platforms such as kiosks and standalone devices, the Web, social networks, consoles, PC/Macs, and mobile phones. The game genres implemented in most of their projects included action or adventure, racing, puzzles, strategy/role playing, sports, music-based, and other categories. The participant organizations distributed the surveys within various departments, and the survey respondents had been employed in that particular organization for at least three years. The survey respondents worked in various capacities from game development to middle and senior management and played a role in either policy-making or implementation of organizational strategies. The total number of survey respondents was 61, including a minimum of two and a maximum of five responses from each organization.

#### 4.1 Measuring instrument

This study gathered data on the key business factors and the perceived level of game performance identified in the research model depicted in Fig. 1. To learn about these two topics, the questionnaire presented in Appendix A was used as a data collection instrument. First, organizations involved in the game development business were asked to what extent they practiced the identified key business factors within their organization. Second, they were asked about the business performance of their games in the digital game industry. The five-point Likert scale was used in the questionnaire, and with

each statement, the respondents were required to specify their level of agreement or disagreement. Thirty-three items were used to measure the independent variables (the key factors), and for the dependent variable (game performance), eight items were used. The literature related to key business factors was reviewed in detail to ensure a comprehensive list of measuring items for each factor from the literature. A multi-item, five-point Likert scale was used to measure the extent to which each key business factor was practiced within the organization. The Likert scale ranged from (1) meaning "strongly disagree" to (5) meaning "strongly agree" and was associated with each item. The items for each identified factor were numbered 1 to 33 in Appendix A and also labelled sequentially. Game business performance was the dependent variable and was measured for at least the past three years in the context of organizational financial strength, market growth, cost savings, and reduced development time based on a multi-item five-point Likert scale. The designated items for the dependent variable were numbered separately from one to eight and labelled sequentially. All the items specifically written for this study are presented in Appendix A. To the best of the authors' knowledge, this is the first empirical study of key digital game business factors in the DGI.

#### 4.2 Reliability and validity analysis

The empirical studies included two integral measures of precision: reliability and validity. Reliability refers to a measurement's reproducibility or consistency, whereas validity refers to the inference or agreement between the true value and the measured value. A reliability and validity analysis was performed for the measuring instrument that was specifically designed for this empirical study. This analysis was based on the most common approaches used in empirical studies. Based on internal consistency analysis, reliability of the multi-scale measurement items for the seven identified factors was evaluated. Cronbach's alpha coefficient [51] was used to evaluate internal consistency. First, Cronbach's alpha coefficient was calculated on a sample dataset which excluded assessment items from each category if they affected the desired value of Cronbach's alpha coefficient. The responses to question 6 on market orientation, question 12 on innovation, question 22 on time to market, question 24 on monetization strategy, and question 33 from brand name strategy were excluded from the investigation based on their effect on Cronbach's alpha coefficient. The reliability analysis for the seven factors is reported in Table 1, with Cronbach's alpha coefficient ranging from 0.61 to 0.76. Satisfactory value criteria for Cronbach's alpha have been reported by a number of researchers based on their findings. Nunnally and Brenste [52] stated that a value of the reliability coefficient of 0.70 or higher for a measuring instrument was satisfactory. Van de Ven and Ferry [53] suggested that a value of 0.55 or higher of the reliability coefficient could be considered satisfactory. Osterhof [54] concluded based on his findings that a value of 0.60 or higher was satisfactory. Hence, all developed variable items for this study could be considered reliable.

Campbell and Fiske [55] concluded that convergent validity has occurred if scale items are highly correlated and in a given assembly, they move in the same direction. For the analysis of convergent validity, principal component analysis (PCA) [56] with seven factors was used, with the results shown in Table 1. The eigenvalues [57] were used as a reference point to determine the construct validity of the PCA-based measuring instrument. For this empirical investigation, the Kaiser Criterion [58] was used, which states that any value greater than one for any component is to be retained. The eigenvalue analysis showed that out of the seven variables, six together formed a single factor, whereas brand name strategy loaded on two factors, and both eigenvalues were greater than one. The reported convergent validity of this study has been considered as adequate.

# Table 1 Cronbach's alpha coefficient and principal component analysis of seven variables

| Business factor         | ltem no. | Coefficient $\alpha$ | PC eigenvalue |
|-------------------------|----------|----------------------|---------------|
| Customer satisfaction   | 1-4      | 0.71                 | 1.49          |
| Market orientation      | 5–8      | 0.67 (Q6)            | 1.57          |
| Innovation              | 9–13     | 0.74 (Q12)           | 1.01          |
| Relationship management | 14–19    | 0.60                 | 1.16          |
| Time to market          | 20–23    | 0.64 (Q22)           | 1.61          |
| Monetization strategy   | 24–28    | 0.61 (Q24)           | 1.25          |
| Brand name strategy     | 29–33    | 0.76 (Q33)           | 1.79          |

#### 10

#### 4.3 Inter-rater agreement analysis

Mostly, there were two or one respondents from one organization. We have performed inter-rater agreement analysis [59] to address the issue of conflicting opinion from same organization. The inter-rater agreement is about the level of agreement in the ratings provided by different respondents for the same process or software engineering practice [60]. Thus, we performed inter-rate agreement analysis to identify the level of agreement among different respondents from same organization. To evaluate inter-rater agreement, Kendall co-efficient of concordance (W) [61] is usually preferred for ordinal data as compared others methods like Cohen's Kappa [62]. "W" represents the difference between the actual agreement drawn from data and perfect agreement. Values of Fleiss Kappa and the Kendall's *W* coefficient can range from 0 (represents complete disagreement) to 1 (represents perfect agreement) [63]. Therefore, Kappa [60] standard includes four levels ranging from: < 0.44 means poor agreement, 0.44 to 0.62 entails moderate agreement, 0.62 to 0.78 indicates substantial agreement, and > 0.78 represents excellent agreement. In this study, the Kappa coefficient observed come under substantial category and ranges from 0.63 to 0.68. Table 2 reports the Kappa and Kendall statistics for five organizations.

#### Table 2

Inter-rater agreement analysis

|               | Kendall's statistics                     |         | Kappa Statistics         |        |
|---------------|--|---------|--------------------------|--------|
| Organizations | Kendall's Coefficient of Concordance (W) | χ2      | Fleiss Kappa Coefficient | Z      |
| А             | 0.72                                     | 58.20*  | 0.68                     | 8.20*  |
| В             | 0.65                                     | 52.90*  | 0.63                     | 7.98** |
| С             | 0.71                                     | 57.42*  | 0.67                     | 8.04*  |
| D             | 0.63                                     | 51.32*  | 0.62                     | 7.54*  |
| E             | 0.74                                     | 60.14** | 0.69                     | 9.01** |

\*Significant at p<0.05

\*\*Insignificant at p>0.05

#### 4.4 Data analysis techniques

Various statistical approaches were used in this research for data analysis. Initially, this activity was split into three phases to estimate the significance of hypotheses H1–H7. Phase I involved parametric statistics and normal distribution tests. In Phase II, partial least squares (PLS) was used as a nonparametric statistical approach. Due to the small sample size, both parametric and nonparametric approaches were used to address the threat to external validity. Multiple items were used in the measuring instrument for each independent variable and the dependent variable, with respondents' ratings for each variable aggregated to obtain a composite value. In phase I, tests were conducted for each hypothesis H1–H7 using parametric statistics such as the Pearson correlation coefficient and the one-tailed *t*-test. In phase II, nonparametric statistics such as the Spearman correlation coefficient were used to test hypotheses H1–H7. In phase III, tests were carried out for research model hypotheses H1–H7 based on the PLS technique. Fornell and Bookstein [64] and Joreskog and Wold [65] reported that if non-normal distribution, complexity, small sample size, and low theoretical information are issues, then PLS will be helpful. The PLS technique was used in Phase III to increase the reliability of the results and deal with the limitation of small sample size. The main reason for the small sample size was first, that most games on the market are developed by one or three developers, but this study targeted game development companies with more than three employees, and second, some companies declined to respond to the survey due to their busy schedule. For statistical calculations, the Minitab 17 software was used.

#### 5. Data Analysis and Results

#### 5.1 Phase I of hypothesis testing

Parametric statistics were used in this phase to test hypotheses H1–H7. The Pearson correlation coefficient was examined between the independent variables (key business factors) and the dependent variable (game performance) of the research model, as illustrated in Figure 1. To accept a hypothesis, the level of significance was selected so that if the *p*-value was



less than 0.05, the hypothesis would be accepted, and if the *p*-value was greater than 0.05, the hypothesis would be rejected [66]. The calculated results for the Pearson correlation coefficient are listed in Table 3. Hypothesis H1 was accepted because the Pearson correlation coefficient for customer satisfaction and game performance was positive (0.50) at p<0.05. For hypothesis H2 concerning market orientation and game performance, the Pearson correlation coefficient was also positive (0.57) at p<0.5, and therefore hypothesis H2 was also accepted. Hypothesis H3 concerning innovation and game performance was rejected due to its higher *p*-value (0.93). Hypothesis H4 concerning relationship management and game performance was also rejected based on its negative Pearson correlation coefficient (-0.361) at p<0.05. Hypothesis H5 concerning time to market and game performance was accepted based on its positive correlation coefficient (0.61) at p<0.05. Hypothesis H6 regarding monetization strategy and game performance was also accepted due to its positive Pearson correlation coefficient (0.25) at p<0.05. The last hypothesis (H7) between brand name strategy and game performance was also found to be significant (0.79) at p<0.05 and was therefore accepted. Hence, in summary, hypotheses H1, H2, H5, H6, and H7 were accepted and found to be statistically significant. Hypotheses H3 and H4 were not supported statistically and were therefore rejected.

| Hypothesis | Key factor              | Pearson correlation coefficient | Spearman correlation coefficient |
|------------|-------------------------|---------------------------------|----------------------------------|
| H1         | Customer satisfaction   | 0.50*                           | 0.55*                            |
| H2         | Market orientation      | 0.57*                           | 0.57*                            |
| H3         | Innovation              | 0.01**                          | 0.13**                           |
| H4         | Relationship management | -0.16**                         | -0.16**                          |
| H5         | Time to market          | 0.61*                           | 0.55*                            |
| H6         | Monetization strategy   | 0.25*                           | 0.27*                            |
| H7         | Brand name strategy     | 0.79*                           | 0.78*                            |

Hypothesis testing using parametric and nonparametric correlation coefficients.

\*Significant at p<0.05

Table 3

\*Insignificant at p>0.05

#### 5.2 Phase II of hypothesis testing

Phase II involved testing hypotheses H1–H7 based on the nonparametric Spearman correlation coefficient. The observations made in this phase for the Spearman correlation coefficient are also reported in Table 2. Hypotheses H1 was accepted because of its positive Spearman correlation coefficient (0.55) at p<0.05. The Spearman correlation coefficient for market orientation and game performance (hypothesis H2) was also positive (0.57) at p<0.05 and was also found to be significant. The relationship between innovation and game performance (hypothesis H3) was not found to be statistically significant due to its Spearman correlation coefficient (0.13) at p>0.05 and was rejected. For hypothesis H4, the Spearman correlation coefficient was negative at p<0.05, and therefore H4 was rejected. Hypothesis H5 concerning time to market and game performance was accepted due to its positive coefficient (0.55) at p<0.05. Hypothesis H6 concerning monetization strategy and game performance was also accepted due to its positive Spearman correlation coefficient (0.27) at p<0.05. The last hypothesis (H7) between brand name strategy and game performance was also found to be significant (0.78) at p<0.0. Hence, in summary, hypotheses H1, H2, H5, H6, and H7 were accepted and found to be statistically significant. Hypotheses H3 and H4 were not supported statistically and were therefore rejected.

#### 5.3 Phase III of hypothesis testing

Phase III included hypothesis testing based on the partial least squares (PLS) technique. PLS was used for cross-validation and to overcome some limitations associated with the results obtained from the parametric and nonparametric statistical approaches used in Phases I and II. Hypotheses H1–H7 were tested for direction and significance. To examine PLS for each hypothesis, the dependent variable (game performance) was designated as the response variable and the individual business factors as the predicate variable. The observed structural test results for the hypotheses are reported in Table 4 and

#### 12

include the observed values of  $R^2$ , the path coefficient, and the *F*-ratio. The path coefficient for customer satisfaction (H1) was observed to be 0.78,  $R^2$  was 0.24, and the *F*-ratio was 19.10, and H1 was found to be significant at p<0.05. Market orientation (H2) had a positive path coefficient of 1.04,  $R^2 = 0.32$ , and *F*-ratio = 28.51 and was also found to be statistically significant at p<0.05. Innovation (H3) had a path coefficient of 0.02, a very low  $R^2$  of 0.0001, and an *F*-ratio of 0.0001 and was found to be insignificant at p<0.05. Relationship management (H4) had a negative path coefficient of -0.27, a low  $R^2$  of 0.01, and an *F*-ratio of 1.69 and was judged to be insignificant because the *p*-value was greater than 0.05. Time to market (H5) (path coefficient: 1.16,  $R^2$ : 0.37, and *F*-ratio: 35.52) had the same direction as proposed. Monetization strategy (H6) (path coefficient: 0.51,  $R^2$ : 0.64, and *F*-ratio: 4.04) and brand name strategy (path coefficient: 0.94,  $R^2$ : 0.62, and *F*-ratio: 100.38) was found significant at p<0.05.

Table 4PLS regression results for hypothesis testing

| Hypothesis | Factors                 | Path coefficient | <b>R</b> <sup>2</sup> | F-Ratio |
|------------|-------------------------|------------------|-----------------------|---------|
| H1         | Customer satisfaction   | 0.78             | 0.24                  | 19.10*  |
| H2         | Market orientation      | 1.04             | 0.32                  | 28.51*  |
| Н3         | Innovation              | 0.02             | 0.01                  | 0.01**  |
| H4         | Relationship management | -0.27            | 0.01                  | 1.69**  |
| H5         | Time to market          | 1.16             | 0.37                  | 35.52*  |
| H6         | Monetization strategy   | 0.51             | 0.64                  | 4.04*   |
| H7         | Brand name strategy     | 0.94             | 0.62                  | 100.38* |

\*Significant at p<0.05

#### 5.4 Research model testing

The linear regression equation for the research model is given by Equation 1. The research model was tested to provide empirical evidence that business factors play a considerable role in digital game performance in the market. The test procedure examined the regression analysis, the model coefficient values, and the direction of the associations. The dependent variable (game performance) was designated as the response variable and the other independent variables (all the key business factors) as predicate variables. The regression analysis model results are reported in Table 5. The path coefficients of five of the seven variables (customer satisfaction, market orientation, time to market, monetization strategy, and brand name strategy) were positive and found to be statistically significant at p<0.05. The path coefficient of innovation was negative and made this factor insignificant in the research model. The overall  $R^2$  value of the research model was 0.74, and the adjusted  $R^2$  value was 0.71 with an *F*-ratio of 21.16, which was significant at p<0.05.

#### Table 5

Linear regression analysis of the research model

| Model coefficient name  | Model coefficient       | Coefficient value       | t-value |  |
|-------------------------|-------------------------|-------------------------|---------|--|
| Customer satisfaction   | $\boldsymbol{\beta}_1$  | 0.23                    | 1.67*   |  |
| Market orientation      | β <sub>2</sub>          | 0.66                    | 3.35*   |  |
| Innovation              | $\boldsymbol{\theta}_3$ | 0.18                    | 1.20**  |  |
| Relationship management | $oldsymbol{eta}_4$      | -0.14                   | -1.05** |  |
| Time to market          | β5                      | 0.02                    | 1.10*   |  |
| Monetization strategy   | $\mathcal{B}_6$         | 0.13                    | 1.68*   |  |
| Brand name strategy     | β7                      | 0.69                    | 4.13*   |  |
| Constant                | $oldsymbol{eta}_o$      | 7.35                    | 2.01*   |  |
| <i>R</i> <sup>2</sup>   | 0.74                    | Adjusted R <sup>2</sup> | 0.71    |  |
| <i>F</i> -ratio         | 21.16*                  |                         |         |  |

#### 6. Discussion

Today's digital era has attracted many people to play games and to develop their own games for profit. Developing a digital game involves activities from different disciplines and has its roots in business management and software engineering. This research aims to help game development organizations understand the interdependencies and relationships between key business factors and game performance in the market. This research offers an opportunity to explore empirically the association between key business factors and digital game performance. This is the first empirical investigation of business factors in relation to game performance, and the results support the theoretical foundations and provide first evidence that key business factors play an important role in digital game performance. This could well result in institutionalizing the digital game production approach in the game development organization, which in turn has a high potential to maximize profits.

Customer satisfaction in the DGI refers to meeting the customer's expectations by providing a functional game, addressing the availability issue for online games, and offering good customer service and expert advice on games. The customer satisfaction variable for business performance measurement in the DGI has not yet been explored in the literature. Basically, game development organizations must value their customers or players by meeting their expectations. This study has found a positive association between customer satisfaction and digital game performance. Organizations can use appropriate measures to track their customers' purchasing behavior and focus more on providing customer service. To implement better customer service, organizations need to understand their game players, implement player-specific platform services, and take feedback strongly into consideration. Most literature reviews have focused on the relationship between business performance and customer satisfaction in different industries. To be successful in the competitive DGI market, game development organizations must take all these strategies into account to explore their relationship with their customers. By adopting best practices, organizations will be able to understand their customers or players, and instead of aiming for one-hit wonders, attracting new customers and retaining existing ones will become the main indicators of customer satisfaction. Important factors affecting customer retention include their initial play experience, the level of game addictiveness, the fit between organizational targets and the market, and finally, the ability of the organization to correct all issue that harm retention. Customer satisfaction data in an organization are also helpful for continuous improvement, which affects the organization's business performance on a long-term basis.

Market orientation was also found to have a positive impact on digital game performance. In the DGI, market orientation is a vast and complex topic. Game development organizations need to focus mainly on two artifacts while developing their games. First, the marketing strategy artifact is a kind of guideline that describes your targets, and second, the marketing plan artifact is a detailed description of your targets and how you will execute them. The organization must develop the marketing strategy at the beginning of the game development process because most of the decisions about game development such as monetization, game design, languages, and demographic locations of game availability will impact the marketing strategy. For market-driven games, one important decision about marketing plan execution will be different. A publisher will take into account the target audience, locations, and platforms, and the marketing plan will be executed by the publisher. However, if the organization publishes the game on their own, its must also consider its target audience, the game business model, geography, budget, platforms, and marketing channels. The impact of market orientation is a significant joint predicator of business performance in terms of return on investment, market share, and profitability. Business performance as a market orientation variable can be measured in terms of monetization, packaging, and promotion strategies as well as calculations of individual customer revenue and profitability.

The DGI appears to consider innovation as a basic source of competiveness. Most organizations see innovation in games as bringing new things to the market and being different from competitors. Innovation in game development can involve application of new ideas at the game level, storyboard production, use of new technology, or the creative artistry of the game, with the aim of attracting more gamers and thus creating value in terms of business performance. Not one single study has addressed the issue of innovation in the DGI. Innovation in the game industry can also refer to an innovative business model of the game development process that addresses all innovation categories, as described by Johannessen [34] and Lawson and Samson [68]. On contrary, the findings of this empirical investigation do not support a statistically positive relation between innovation and digital game performance. The direction of association was positive, but the required statistical level of confidence was not supported.

It has also been assumed here that the user integration approach in the DGI enables organizations to use their users as a resource. It is important to consider users as a resource because especially in the computer game industry, users are the revenue producer, and the business totally depends on their positive playing experience. More user involvement enables the organization to retain its users/customers. The question now arises of how game users who are also players can become involved in parts of the game development process. One way of user integration is through virtual community membership. Nohria and Ghoshal [69] argue that "the real leverage lies in creating a shared context and common purpose and in enhancing the communication densities within and across the organization's internal and external boundaries". This argument also supports the concepts of customer socialization and community participation in the game development process. However, this user integration approach is cost-effective for any organization. In such communities, customers can participate based on their broad communities of interest. They can be a part of game development by sharing their playing experiences, being involved in idea generation, becoming co-creators or testers of games, or in other ways. Use of online communities in the development process constitutes an important source of innovation and also enables organizations to implement constructive relationships with their users.

In successful game development, relationship management plays a significant role. Integrating players into the development process and maintaining excellent working relationships with them helps developers to improve the performance and functionalities of their games. However, the assumption that relationship management also helps the organization to understand its customers' needs better and remain up-to-date about market trends was not found to be significant in this study. Empirical investigation found a negative association and also insufficient statistical support for a significant confidence level. Hence, the study was not able to find any impact or association between relationship management and digital game performance.

Because the DGI is flourishing, competition is very tough between digital game organizations. The organization which achieves competitive advantage using time-to-market processes will have a positive impact on business performance. This hypothesis was also supported by empirical investigation. Hence, game launch timing is important to capture major market share. The time-to-market approach in a game development organization develops a publishing schedule for the game and provides essential guidelines for development schedules to the developers. The game launch schedule is a crucial business decision that has profound and long-lasting impact on the business performance of an organization in retaining and capturing the market.

In the DGI, fulfillment of financial objectives or monetization strategy depends on economically optimizing the pricing scheme for customers, the cost structure, and the target customer segment. In this empirical investigation, a positive association was found between monetization strategy and digital game performance. The pricing scheme can be a one-time payment, pay per session, pay per play, or subscription-based or bundled pricing. The cost structure is based on the complete picture of the entire budget for game development, including marketing and distribution costs. The overall cost of each phase until delivery to the user directly impacts the overall profitability of the organization. However, it is difficult to measure the cost impact of each phase on overall business performance [46]. In this situation, the impact of monetization can be measured by using the overall profitability of the organization as a measure of business performance. As for the target customer segment, it is important to understand the needs of target customer groups to ensure that games are properly priced, marketed, and packaged to achieve business success.

Recently, in the DGI, use of successful game development brands that are useful to particular market segments has helped organizations connect with their target audience. This empirical investigation found a positive association and impact of brand name strategy on overall game performance. In particular, brand name strategy has become marketing strategy in branded games. Although branded game development costs the organization more, it pays off after publication by attracting large numbers of new and repeat users. An effective brand name strategy helps in defining game development and execution, ensuring that the game gets appropriate promotion in the marketplace, and positioning the game for its target audience.

Overall, the findings of the study are important for the development of good quality digital game. Rapid and continual changes in technology and intense competition not only affect the business, but also have a great impact on development activities. To deal with this strong competition and high pressure, game development organizations must continuously assess their activities and adopt a proper evaluation methodology. Use of a proper assessment methodology will help the organization identify its strengths and weaknesses and provide guidance for improvement. However, the fragmented nature of the game development process requires a comprehensive evaluation strategy which has not yet been entirely explored.

14

#### 6.1 Limitations and Threats to External Validity

Metrics, surveys, case studies, and experiments are some examples of empirical techniques used for software engineering processes and product investigations. However, certain limitations are associated with these empirical investigations and with this study as well.

The first limitation of this study was the selection and choice of independent variables. Seven independent variables were included to investigate their association with and impact on digital game performance. However, other key factors may exist that have a positive impact on digital game performance, but this study was limited to the seven variables because of their presence in the literature. In addition, other key factors may exist, such as environmentally based, regionally based, or political factors, which have a positive impact on digital game performance, but are not considered in this study. Furthermore, this study has focused only on business factors in digital game performance.

The second notable limitation of the study is the small sample size. Most game industry developers who follow either agile practices or poor development practices were unable to respond to the questionnaire and did not respond. The vast majority of game developers work in one- to three-person teams and did not have the required level of experience (three years) and were therefore excluded from this empirical investigation. Most respondents refused to answer the questionnaire because they were too busy in the game development process or launching their games in the market. Some game development organizations are also hesitant to disclose their business performance. Therefore, data collection from the game industry was limited, resulting in small sample size. The number of respondents from one organization was beyond the authors' control because the organization's upper management was responsible for distributing the survey within a company. The main effect of small sample size is on its statistical power, Type II error, significance and on distribution [70]. Therefore, the important thing is while making conclusion avoid strong statements. As, the small samples size studies results can be difficult to replicate or generalize [71] but they do provide some interplay between variables. The well designed small studies are seems ok to conduct as they provide quick results but they need to be interpreted carefully [72]. The low sample size constraint of this study makes the results difficult to generalize. However, the results of this study are useful in providing some basic foundation to design larger confirmatory study, which is the future objective of this work.

Biased decision-making was the third limitation of this study. Although multiple responses were collected from each company to address the bias issue, but it remained a core issue. Respondents were asked to consult available documentation within a company to fill out the survey. Accepted psychometric principles were used to design the assessment items, for conflicting opinions from same organization inter-rater agreement analysis was performed but the measuring instrument was still based on individual subjective assessment.

In spite of its specific and general limitations, this study has contributed to the field of digital games and has helped game development organizations to understand the business dimension of digital games.

#### 7. Conclusions

Game development is an interdisciplinary concept that embraces software engineering, business, management, and artistic disciplines. This research has facilitated a better understanding of the business dimension of digital games. The main objective of this research was to investigate empirically the effect of business factors on the performance of digital games in the market and to try to find answers to the research questions posed in this study. Empirical investigation results demonstrated that business factors play an important role in digital game performance. The results of the study strongly indicate that customer satisfaction, time to market, monetization strategy, market orientation, and brand name strategy are positively associated with the performance of a digital game organization. The empirical investigation found no strong association or impact between relationship management or innovation and digital game performance.

#### 16

This study is the first of its kind in the field of digital games. It will help and enable organizations to achieve a better understanding of the effectiveness of business factors and their role in terms of game performance in the market. Game development organizations need to consider these various business factors over and above their current efforts to improve the performance of their developed games in the market.

Currently, the authors are working on developing a digital game maturity model for game development process assessment. This study has provided the empirical evidence and justification to include business factors in evaluating the business dimension of game development process maturity.

214

#### Appendix A: Key business factors measuring instrument

| S | E( | CTI | ON | ONE |   |
|---|----|-----|----|-----|---|
| - | -  | _   |    |     | - |

| 1.1 | <b>Participant</b> | <u>details</u> |
|-----|--------------------|----------------|
|     |                    |                |

| Full Name (Optional)     | Job Title/Position |
|--------------------------|--------------------|
| _                        |                    |
| Experience (in years)    |                    |
|                          |                    |
| Address                  |                    |
|                          |                    |
| Telephone no. (optional) |                    |
|                          |                    |
| Email                    |                    |
|                          |                    |

#### 1.2 Demographics

| Country in which the company is located?  |
|---|
| Please Specify:   |
|   |
| What is the scope of your company?  |
|   |
| National Multinational Don't Know   |
| Please Specify:   |
|   |
| Approximately how many people are employed by your company? ( Please tick the appropriate box)            |
|   |
| Less than 20 20-70 More than 100 Not sure   |
|   |
| Please Specify:   |
|   |
| What type of game genre is developed by your company and what is the target platform for developed games? |
| Please Specify:   |
|   |
| Who are the target audience?  |
| Please Specify:   |
|   |

#### SECTION TWO

#### 2.1 Evaluation of business performance success factors identified through literature review

The questionnaire objective is to find out which factors have a positive impact on business performance. Please select the correct scale based on your best knowledge.

|       | Business performance key factors for game development companies  |   |   |          |   |   |     |
|-------|--|---|---|----------|---|---|-----|
| Liker | scale (1 = strongly disagree; 2= disagree; 3 = neutral; 4= agree; 5 strongly agree)                              | 1 | 2 | 3        | 4 | 5 | N/A |
| Custo | mer or player satisfaction   |   |   |          |   |   |     |
| 1     | The organization is using a game rating scheme to respond to player requests.                                    |   |   |          |   |   |     |
| 2     | The organization has a good customer service department.   |   |   |          |   |   |     |
| 3     | The organization provides expert advice on games.  |   |   |          |   |   |     |
| 4     | The organization provides feedback or response to its customers.   |   |   |          |   |   | )   |
| Mark  | et orientation   |   |   |          |   |   |     |
| 5     | The organization has adequate skills and resource to perform detailed market studies to determine what           |   |   |          |   |   |     |
|       | types of games are in demand and who will be the target audience.  |   |   |          |   |   |     |
| 6     | The organization uses appropriate feedback mechanisms to ensure game quality.                                    |   |   |          |   |   |     |
| 7     | The organization always develops a proper marketing plan and strategy to gain competitive advantage.             |   |   |          |   |   |     |
| 8     | The organization is able to maximize market size and its growth over time.                                       |   | Ť |          |   |   |     |
| nnov  | ration   |   |   |          |   |   |     |
| 9     | The organization is able to use innovative ideas successfully for game development and game level repositioning. |   |   |          |   |   |     |
| 10    | The innovations in games are aligned with existing business goals.   |   |   |          |   |   |     |
| 11    | Reactive and proactive innovation in the game development process is supported by management.                    |   | l | l        |   |   |     |
| 12    | Past innovative measures taken by the organization have helped in improving the game development and             |   | l | l        |   |   |     |
|       | management process.  |   |   |          |   |   |     |
| 13    | The organization believes that R&D investment can yield positive results in the near future.                     |   |   |          |   |   |     |
|       | ionship management   |   |   |          |   |   |     |
| 14    | The organization has well-established mechanisms for data extraction, manipulation, and production for           |   |   |          |   |   |     |
|       | customer profiling, profitability analysis, and retention modeling.  |   |   |          |   |   |     |
| 15    | The organization participates in online gaming communities to identify player concerns.                          |   |   |          |   |   |     |
| 16    | The organization is able to retain players for long periods.   |   |   |          |   |   |     |
| 17    | The organization has established a balanced player- and game-centered strategy for game development.             |   |   |          |   |   |     |
| 18    | The organization is able to attract new players and retain existing ones using innovative targeted methods       |   |   |          |   |   |     |
|       | and personalized communication.  |   |   |          |   |   |     |
| 19    | The organization is using a user integration strategy for game development.                                      |   |   |          |   |   |     |
|       | to market  |   |   |          |   |   |     |
| 20    | Games are launched in the market before competitors' games.  |   |   |          |   |   |     |
| 21    | The organization regularly studies and researches development updates, market reviews, and game                  |   |   |          |   |   |     |
|       | publishing schedules to build awareness of market needs and trends.  |   |   |          |   |   |     |
| 22    | The organization publishes games in response to competitors' actions.  |   |   |          |   |   |     |
| 23    | Being first in the market helps to retain players and tends to attract new ones.                                 |   |   |          |   |   |     |
|       | etization strategy   |   |   |          |   |   |     |
| 24    | The organization is able to achieve its financial objectives successfully.                                       |   |   |          |   |   |     |
| 25    | The organization is able to use cost-saving strategies successfully.   |   |   |          |   |   |     |
| 26    | The organization is able to acquire more players for less investment.  |   |   |          |   |   |     |
| 27    | The organization uses in-depth mechanics to maximize conversion rate and lifetime value in games.                |   |   |          |   |   |     |
| 28    | The organization can successfully build cross-platform offerings to reach players/consumers.                     |   |   |          |   |   |     |
| _     | name strategy  |   |   |          |   |   |     |
| 29    | The game development process of the organization is unique and different from its competitors in the             |   |   |          |   |   |     |
|       | market.  |   |   |          |   |   |     |
| 30    | New games and their latest versions are consistent with brand extensions.  |   |   |          |   |   |     |
| 31    | The latest game or its extended version attracts new customers and retains existing one because it is            |   |   |          |   |   |     |
|       | considered an improvement in a newer or existing game.   |   |   |          |   |   |     |
| 32    | The buying decision of the customer is based on brand name loyalty.  |   | ŀ | ŀ        |   |   |     |
| 33    | Published games have one-to-one competition in the market.   |   |   |          |   |   |     |
|       | e business performance   |   | - | -        | - |   |     |
| 1     | The organization was able to reduce the development time and cost of games over the last five years.             |   | - | -        |   |   |     |
| 2     |  |   | - | -        |   |   |     |
| ۷     | The organization's sales have improved gradually over the last five years.                                       |   |   | <u> </u> | - |   |     |
| 3     | The organization's financial analysis shows progressive growth over the last five years.                         |   |   |          |   |   |     |

| 5 | The organization has been able to reduce significantly the number of competitors over the last five years. |  |  |  |
|---|--|--|--|--|
| 6 | The organization is considered as a pioneer in the digital game industry rather than as a follower.        |  |  |  |
| 7 | Customer satisfaction and loyalty ratings have increased over the last five years.                         |  |  |  |
| 8 | The business goals of the organization have been successfully accomplished.                                |  |  |  |

#### References

- [1] The New York Times Bits Blog, July 5, 2011. Video game industry continues major growth. Gartner Says. Accessed on 25<sup>th</sup> October, 2014.
- PWC global media and entertainment outlook 2011–2014, 2011. Available at http://www.pwc.com/gx/en/global-entertainment-mediaoutlook/territory-segments-digital-forecast-overview.jhtml. Accessed on July 28, 2013.
- [3] Haukka, S., 2011. Working in Australia's digital games industry. Consolidation Study. Brisbane: Queensland University of Technology. Australian Research Council and Games Developers' Association of Australia.
- [4] The games sector in Ireland: An action plan for growth. Available at http://www.forfas.ie/media/forfas20111010-Games\_Sector\_in\_Ireland.pdf. Accessed on July 27, 2013.
- [5] Kerr, A. 2000. Ireland in the global information economy: Innovation and multimedia 'Content' industries. Javnost the public, 7.
- [6] Levitt, T., 1981. Marketing Intangible Products and Product Intangibles, Harvard Business Review. Marketing May Issue.
- [7] Burnham, V., 2003. Supercade: A visual history of the video game age 1971-1984, The MIT press.
- [8] Sellers, J., 2001. Arcade fever: The fans guide to the golden age of video.
- [9] Rabin, S., 2010. Intoduction to game development, 2<sup>nd</sup> Edition, Course technology, CENGAGE Technology.
- [10] Neogames center of game business, 2011. Research and Development. The Finnish Game Industry Report.
- [11] Oliver, R. L., 1997. Satisfaction: A behavioral perspective on the consumer. New York: McGraw-Hill.
- [12] Zeithaml, V. A., 2000. Service quality, profitability, and the economic worth of customers: what we know and what we need to learn. Journal of the Academy of Marketing Science, 28(1), 67–85.
- [13] Koska, M. T., 1990. High-quality care and hospital profits: is there a link? Journal of Hospitals, 62(3).
- [14] Nelson, E., Rust, R.T., Zahorik, A., Rose, R. L., Batalden, P. and Siemanski, B., 1992. Do patient perceptions of quality relate to hospital financial performance? Journal of Healthcare Marketing, December, 1–13.
- [15] Aaker, D. A. and Jacobson, R., 1994. The financial information content of perceived quality. Journal of Marketing, 58, 191-201.
- [16] Anderson, E. W., Fornell, C. and Lehmann, D. R., 1994. Customer satisfaction and word of mouth. Journal of Service Marketing, 1(1), 5–17.
- [17] Ittner, C. and Larckner, D. F., 1996. Measuring the impact of quality initiatives on firm financial performance. In: S. Ghosh and D. Fedor (Eds.), Advances in the Management of Organisational Quality. JAI, 1, 1–37.
- [18] Fornell, C., 1992. A national satisfaction barometer: the Swedish experience. Journal of Marketing, 56, 6–21.
- [19] Hallowell, R., 1996. The relationship of customer satisfaction, customer loyalty, and profitability: an empirical study. International Journal of Service Industry Management, 7(4), 27–42.
- [20] Zhang, C. and Pan, F., 2009. The impacts of customer satisfaction on profitability: a study of state-owned enterprises in China. Journal of Service Science, 1(1), 21–30.
- [21] Johnson, L., 2001. Using the critical incident technique to assess gaming customer satisfaction. UNLV Gaming Research & Review Journal, 6(2), 1– 12.
- [22] Lu, H. and Wang, S., 2008. The role of Internet addiction in online game loyalty: an exploratory study. Journal of Internet Research, 18(5), 499-519.
- [23] Van der Wiele, T., Boselie, P. and Hesselink, M., 2001. Empirical evidence for the relation between customer satisfaction and business performance? In: QMOD'2001 Conference, Linköping, Sweden, September 12–14, pp. 1–23.
- [24] Zeithaml, C. P. and Zeithaml, V. A., 1984. Environmental management: revising the marketing perspective. Journal of Marketing, 48(2), 46-53.
- [25] Hunt, D. S., 2004. On the services-centered dominant logic for marketing. Journal of Marketing, 68(1), 21-22.
- [26] Berry, L. L., 1983. Relationship marketing. In: L. L. Berry, G. L. Shostack, and G. Upah (Eds.), Emerging perspectives on services marketing, Chicago: American Marketing Association, 25–28.
- [27] Gronroos, C., 1990. Service management and marketing: Managing the moments of truth in service competition. Free Press/ Lexington Books, Lexington, MA.
- [28] Fornell, C. and Wernerfelt, B., 1987. Defensive marketing strategy by customer complaint management: a theoretical analysis. Journal of Marketing Research, 24, 337–346.
- [29] Owomoyela, S. K., Oyeniyi, K. O. and Ola, O. S., 2013. Investigating the impact of marketing mix elements on consumer loyalty: an empirical study of Nigerian Breweries PLC. Interdisciplinary Journal of Contemporary Research in Business, 4(11), 485–496.
- [30] Lee, S. H., Lee, G. H., Cho, H. H., Song, D. H. and Rhew, S. Y., 2006. An empirical model of the game software development process. In: Fourth International Conference on Software Engineering Research, Management, and Applications, IEEE, Seattle, Washington, USA, pp. 371–377.
- [31] Katsaliaki, K. and Mustafee, N., 2012. A survey of serious games on sustainable development. In: 2012 Winter Simulation Conference, IEEE Computer Society, pp. 1528-1540.
- [32] Xin, C., 2008. Influence from serious games on mobile game developers' commercial strategies. In: International Seminar on Business and Information Management, IEEE, pp. 207–209.

- [33] Afuah, A., 2009. Strategic innovation: New game strategies for competitive advantage. Routledge: New York.
- [34] Johannessen, J. A., 2013. Innovation: a systematic perspective—developing a systemic innovation theory. Journal of Kybernetes, 42(8), 1195–1217.
- [35] Comviva Technologies Ltd., 2009. Realizing the potential of mobile gaming. White Paper.
- [36] Lindgardth, Z., Reeves, M., Stalk, G. and Deimler, M. S., 2009. Business model innovation. Boston Consulting Group Report.
- [37] Wilson, D. T., 1995. An integrated model of buyer-seller relationships. Journal of the Academy of Marketing Science, 23, 335–345.
- [38] Plé, L., Lecocq, X. and Angot, J., 2008. Customer-Integrated business models: A theoretical framework. Lille Economie and Management (LEM) UMR 8179, IÉSEG School of Management, IAE de Lille, USTL.
- [39] Lecocq, X., Demil, B. and Warnier, V., 2006. Le business model, un outil d'analyse stratégique. L'Expansion Management Review, 123, 96-109.
- [40] Stanely, K. G., Pinelle, D., Bandurka, A., McDine, D. and Mandryk, R. L., 2008. Integrating cumulative context into computer games. In: International Conference on Future Play, ACM, Toronto, Canada, November 3–5, pp. 248–251.
- [41] Ermi, L. and Mayra, F., 2005. Fundamental components of the gameplay experience: analyzing immersion. In: Proceedings of DiGRA 2005: Changing Views: World in Play. Available at http://www.digra.org/dl/db/06276.41516.pdf (Accessed July 14, 2014).
- [42] Ansoff, H. and Stewart, J., 1967. Strategies for technology-based business. Harvard Business Review, 43, 71-83.
- [43] Robinson, W., Fornell, C. and Sullivan, M., 1992. Are market pioneers intrinsically better than later entrants? Journal of Strategic Management, 13, 609–624.
- [44] Ramadan, R. and Widyani, Y., 2013. Game development life cycle guidelines. In: 5<sup>th</sup> International Conference on Advanced Computer Science and Information Systems (ICACIS), IEEE, Jakarta, Indonesia, September 28–29, pp. 95–100.
- [45] Business Builders, 2004. Corporate objective and performance measurement examples. Available at: http://www.businessbuilders.bz/Objective&Performance-Measurement-Examples.pdf.
- [46] Zulkiffli, S. and Perera, N., 2011. A literature analysis on business performance for SMES—subjective or objective measures? In: 2011 SIBR Conference on Interdisciplinary Business and Economics Research, Bangkok, Thailand; Society of Interdisciplinary Business Research (SIBR), pp. 1–9.
- [47] Bennett, P. D., 1988. Dictionary of marketing terms. American Marketing Association.
- [48] Bergstrom, A., 2004. Cyber branding: leveraging your brand on the Internet. Strategy and Leadership, 28 (4), 10–15.
- [49] Davenport, T. H., 1993. Process innovation. Harvard Business School Press: Boston, MA.
- [50] Aguilar-Sav'en, R. S., 2004. Business process modeling: review and framework. International Journal of Production Economics, 90, 129-149.
- [51] Cronbach, L. J., 1951. Coefficient alpha and internal consistency of tests. Psychometrica, 16, 297-334.
- [52] Nunnally, J. C. and Bernste, I. A., 1994. Psychometric Theory. McGraw-Hill.
- [53] Van de Ven, A. H. and Ferry, D. L., 1980. Measuring and assessing organizations. Wiley: New York.
- [54] Osterhof, A., 2001. Classroom applications of educational measurement. Prentice-Hall: New Jersey.
- [55] Campbell, D. T. and Fiske, D. W., 1959. Convergent and discriminant validation by the multi-trait multi-method matrix. Psychological Bulletin, 56(2), 81–105.
- [56] Comrey, A. L. and Lee, H. B., 1992. A first course on factor analysis, 2nd ed. Hillsdale, NJ: Lawrence Erlbaum Associates.
- [57] Kaiser, H. F., 1970. A second generation little jiffy. Psychometrika, 35, 401–417.
- [58] Kaiser, H. F., 1960. The application of electronic computers to factor analysis. Educational and Psychological Measurement, 20, 141–151.
- [59] Landis, J. and Koch, G.G., 1997. The measurement of observer agreement for categorical data, Biometrics, 33, 159-174.
- [60] Emam, K. EL., 1999. Benchmarking kappa: inter-rater agreement in software process assessments, Empirical Software Engineering 4 (2), 113-133.
- [61] von Eye, A. and Mun, E.Y., 2005. Analyzing Rater Agreement Manifest Variable Methods, LEA Publishers, London.
- [62] Cohen, J., 1960. A coefficient of agreement for nominal scales, Educational and Psychological Measurement, 20, 37-46.
- [63] Lee, H.Y., Jung, H. W., Chung, C.S., Lee, J.M., Lee, K.W., and Jeong, H. J., 2001. Analysis of inter-rater agreement in ISO/IEC 15504-based software process assessment, In: Proceedings of the 2nd Asia-Pacific Conference on Quality Software, pp. 341-348.
- [64] Fornell, C. and Bookstein, F. L., 1982. Two structural equation models: LISREL and PLS applied to consumer exit voice theory. Journal of Marketing Research, 19, 440–452.
- [65] Joreskog, K. and Wold, H., 1982. Systems under indirect observation: Causality, structure, and prediction. North Holland.
- [66] Detail of P-value. Available at http://en.wikipedia.org/wiki/P-value, accessed on August 30, 2014.
- [67] Adewale, G., Adesola, M. A. and Oyewale, I. O., 2013. Impact of marketing strategy on business performance: a study of selected small and medium enterprises (SEMS) in Oluyle local government, Ibadan, Nigeria. Journal of Business and Management, 11(4), 59–66.
- [68] Lawson, B. and Samson, D. 2001. Developing innovation capability in organizations: a dynamic capabilities approach. International Journal of Innovation Management, 5(3), 377–400.
- [69] Nohria, N. and Ghoshal, S., 1997. The differentiated network: Organizing multinational corporations for value creation. San Francisco: Jossey-Bass
- [70] Chow, S., Shao, J. and Wang, H., 2008. Sample size calculations in clinical research, 2<sup>nd</sup> Edition, Chapman & Hall CRC, Tylaor & Francis.
- [71] Bryman, A., 2008. Social research methods. Oxford: Oxford University Press.
- [72] Heckshaw, A. 2008. Small Studies: Strength and limitations, European Respiratory Journal, 32 (5), 1141-1143.

CRIF

#### **Article Highlights:**

We reviewed the literature of business factors especially for digital game development organization.

We identified seven key business factors for digital game development organization

The study empirically investigate the significance of business key factors for game development organization.